Author’s response to reviews

Title: Isolation, characterization and antimicrobial evaluation of a novel compound N-octacosan 7beta ol, from Fumaria parviflora Lam.

Authors:

MOHAMMAD JAMEEL (ahmadiamil84@gmail.com)
MOHAMMAD ISLAMUDDIN (mdislam2009@gmail.com)
ABUZER ALI (abuzerali007@gmail.com)
FARHAT AFRIN DR (afрин farhat@yahoo.co.in)
MOHAMMED ALI PROF (maliphyto@gmail.com)

Version: 2 Date: 10 January 2014

Author’s response to reviews: see over
Date: 09/01/2014

To,
The Editor
BMC Complementary and Alternative Medicine

Respected sir,

As per precious and kind suggestions by the experts we are re-submitting the revise manuscript (MS: 4561549721126702) entitled as “Isolation, characterization and antimicrobial evaluation of a novel compound N-octacosan 7β ol, from Fumaria parviflora Lam.” to your esteemed journal for publication.

Your kind gesture will definitely encourage to us.

Thanking you with anticipation

Sincerely yours

PROF. MOHAMMED ALI
Department of Pharmacognosy and Phytochemistry,
Jamia Hamdard
Hamdard University
New Delhi- 110 062
INDIA
maliphyto@gmail.com